Author's response to reviews

Title: Study of the Outcome of Suicide Attempts and the Potential Needs of the Individuals Involved

Authors:

Kaoru Kudo (kudok@iwate-med.ac.jp)
Kotaro Otsuka (kotaro29@df6.so-net.ne.jp)
Jin Endo (aak49040@yahoo.co.jp)
Tomoyuki Yoshida (tomyoshi@iwate-med.ac.jp)
Hisayasu Isono (isono-h@hotmail.co.jp)
Takehito Yambe (ta-yambe@mpd.biglobe.ne.jp)
Hikaru Nakamura (odessa@mub.biglobe.ne.jp)
Sachiyko Kawamura (sakawamu@iwate-med.ac.jp)
Atsuhiko Koeda (mcdaniel.3c.player@hotmail.co.jp)
Junko Yagi (yagiji@flamenco.plala.or.jp)
Nobuo Kemuyama (kemu-von-zibu@live.jp)
Hisako Harada (FZH02260@nifty.com)
Fuminori Chida (chida-psy@umin.net)
Shigeatsu Endo (sendo@iwate-med.ac.jp)
Akio Sakai (sakaiaki@iwate-med.ac.jp)

Version: 2 Date: 17 October 2009

Author's response to reviews: see over
Dear Editors-in-Chief
of BMC Psychiatry

We like to submit the revised version paper entitled “Study of the Outcome of Suicide Attempts and the Potential Needs of the Individuals Involved.” We added the references to this manuscript.
I really appreciate it if you consider to publish our paper mentioned above in your journal.

Very cordially,

Correspondence author:
Kotaro Otsuka, MD, PhD
Dept. of Neuropsychiatry, Iwate Medical University
Adress: 19-1, Uchimaru, Morioka, Iwate, 020-8505, Japan
Phone: +81-19-651-5111
Fax: +81-19-626-4807
E-mail: kotaro29@df6.so-net.ne.jp